Author's response to reviews

Title: Clinical evaluation of a modified silver fluoride application technique designed to facilitate lesion assessment in outreach programs.

Authors:

Graham G Craig (dentaloutlook@bigpond.com)  
Keith R Powell (kp155@iprimus.com.au)

Version: 2 Date: 31 July 2013

Author's response to reviews:

Addition to the original covering letter:
The ethical aspects of the study were evaluated and approved by the Principal Dental Officer and his advisers at the New South Wales Department of Health, Orana and Far West Region.

The last author of the paper inadvertently omitted in the on-line submission has been added.